Author's response to reviews

Title: The frequent BRCA1 mutation 1135insA has multiple origins: a haplotype study in different populations

Authors:

Teresa M Rudkin (teresa@mrs.mni.mcgill.ca)
Nancy Hamel (nancy.hamel@mail.mcgill.ca)
Maria Galvez (maria.galvez@mail.mcgill.ca)
Frans Hogervorst (f.hogervorst@nki.nl)
Johan JP Gille (jig.gille@vumc.nl)
Pal Moller (pal.moller@medisin.uio.no)
Jaran Apold (jaran.apold@helse-bergen.no)
William D Foulkes (william.foulkes@staff.mcgill.ca)

Version: 5  Date: 31 January 2006

Author's response to reviews:

The only content change to the manuscript was to add to the methods section the following sentence:
"Appropriate institutional informed consent guidelines were followed for all recruited patients."

Other changes involved formatting.